Author's response to reviews

Title: Osteoblastic flare in a patient with advanced gastric cancer after treatment with pemetrexed and oxaliplatin: implications for response assessment with RECIST criteria.

Authors:

Vito Amoroso (vitoamoroso@alice.it)
Frida Pittiani (f.pittiani@sirm.org)
Salvatore Grisanti (salvatore.grisanti@tin.it)
Francesca Valcamonico (franzval@yahoo.it)
Edda Simoncini (edda.simoncini@virgilio.it)
Vittorio Ferrari (vittoriodf59@excite.it)
Giovanni Marini (bscivile@numerica.it)

Version: 3 Date: 26 May 2007

Author's response to reviews: see over
To BioMed Central Editorial Production Team

I submit to your kind attention the final version of the manuscript: 6431036821135487
Osteoblastic flare in a patient with advanced gastric cancer after treatment with pemetrexed and oxaliplatin: implications for response assessment with RECIST criteria.

I have made the following changes to the format of the manuscript:

1. Unnecessary capitalisation were removed in headline.
2. The subtitle “conclusion” was modified in “conclusions”.
3. References were placed before the full stop.
4. The two figures were merged in a single composite illustration which contains all parts of the figure sublabelled A and B.

I confirm the acceptance of processing charge, if the article is published.

With my best regards,

Vito Amoroso

-----------------------------------------------------------------------------------------------------------------

Vito Amoroso, MD
Medical Oncology Unit
Azienda Spedali Civili
25100 – Brescia
Italy
Phone: +39-030-3995260
Fax: +39-030-3700017
e-mail: vitoamoroso@alice.it